Author’s response to reviews

Title: The therapeutic value of cerebrospinal fluid ctDNA detection by Next-Generation Sequencing for meningeal carcinomatosis: a case report

Authors:
Xiaosu Guo (guoxiaosu1@163.com)
Junzhao Cui (1097709288@qq.com)
Yue Zhao (1727945768@qq.com)
Weixin Han (862624896@qq.com)
Yueli Zou (sunnygirlzyl@126.com)
Ruiping Gao (310691385@qq.com)
Qing Li (2362797612@qq.com)
Xiaoqing Li (337681974@qq.com)
Junying He (hjy_zn@126.com)
Hui Bu (buhui881@163.com)

Version: 5 Date: 19 Feb 2019

Author’s response to reviews:

Dear Shona Reeves,

We are submitting the revised manuscript with the ID NURL-D-18-00509R4, entitled "The therapeutic value of cerebrospinal fluid ctDNA detection by Next-Generation Sequencing for meningeal carcinomatosis: a case report" for publication in BMC Neurology. According to the comments, we have already revised the manuscript. The following are our reply to the comments, item by item.

Once again, thank you for your kind reconsideration.

Yours sincerely,

Hui Bu
Comments:

1. Manuscript Headings
In order to be in line with journal requirements for Case Reports, please reformat your manuscript using the following headings: "Background", "Case Presentation", and "Discussion and Conclusions". We recommend you move the information currently in your "Methods" and "Results" sections to the "Case Presentation" section. For further information about what should be included in each section please see the submission guidelines: https://bmcneurol.biomedcentral.com/submission-guidelines/preparing-your-manuscript/case-report
Answer: We thank the Reviewer for his/her kind evaluation of our work. As suggested, we have reformatted your manuscript. Thanks!

2. Clean Manuscript
Thank you for uploading a clean version of your manuscript, please again upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please also ensure that all figures, tables, and supplementary/additional files are cited in the text.
Answer: We thank the Reviewer for his/her kind evaluation of our work. As suggested, we have uploaded our manuscript and Figures as the single, final, clean versions. Thanks!